Author's response to reviews

Title: Goodpasture's syndrome with positive C-ANCA and normal renal function - a case report

Authors:

Arunachalam Ramaswami (ramsdr@yahoo.com)
Thiraviam kandaswamy (tknephro@yahoo.com)
Tholappan Rajendran (ammukkannu@yahoo.com)
Hla Aung (hlaaung@brunet.bn)
chakko K Jacob (ckjacob@hotmail.co.uk)
shaukat Zinna (zinnashaukat@yahoo.com)
pemasiri upali Telesinge (ptelesinge@yahoo.co.sg)

Version: 2 Date: 19 December 2007

Author's response to reviews:

Dear sirs, All the suggestions made by you have been done in this revised manuscript. thanking you, yours truly Dr.A.Ramasawami